Author's response to reviews

**Title:** Effectiveness of psychotherapeutic, pharmacological, and combined treatments for chronic depression: a systematic review (METACHRON)

**Authors:**

Levente Kriston (l.kriston@uke.uni-hamburg.de)  
Alessa von Wolff (a.von-wolff@uke.uni-hamburg.de)  
Lars Hölzel (lars.hoelzel@uniklinik-freiburg.de)

**Version:** 3  
**Date:** 23 November 2010

**Author's response to reviews:** see over
Dear Editor,

thank you for considering our manuscript for publication.

Please find a point-by-point response to the formatting requests below.

**Major revisions (we require the author to make these changes)**

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors’ contributions; Authors’ information (optional); Acknowledgements (optional); References

We have restructured the manuscript according to the suggested headings. In order to do this, two minor changes were needed. First, without changing the content of the text, one paragraph was moved from the Background into the Discussion section. Second, a third level of headings (a second-level subheading in bold italics) had to be introduced. We believe that these changes do not represent any substantial modification to the content of the manuscript.
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Affiliations: Please remember to include the country in the affiliation details

*Country information has now been added.*

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

*Tracking has been removed (all changes accepted).*

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

*Typography has been checked.*

Please do not hesitate to contact me if you have any questions.

With kindest regards,

[Signature]

Levente Kriston
(on behalf of all authors)